Author's response to reviews

Title: Brazilian pregnant and lactating women do not change their food intake to meet nutritional goals.

Authors:

Quenia Santos (quenia1104@gmail.com)
Rosely Sichieri (rosely.sichieri@gmail.com)
Dirce ML Marchioni (dirce.marchioni@gmail.com)
Eliseu Verly -Junior (eliseujunior@gmail.com)

Version: 4
Date: 14 May 2014

Author's response to reviews: see over
Date: May 14, 2014  
To: Editor-in-Chief  
From: "Quenia dos Santos"  
Subject: Cover Letter – Response to editor 2

Manuscript: “Brazilian pregnant and lactating women do not change their food intake to meet nutritional goals.”

Corrections suggested by editor:

1- On page 3, line 59, change “10 years older” by “ten years and older.”  
   Answer: It was changed.

2- On page 4, line 60; remove the word “in”.  
   Answer: It was removed.

3- On page 4, lines 62 and 63, remove the word “at” and include the words “within” and “range”.  
   Answer: The suggestions were accepted and the changes were made.

4- On page 4, lines 75 to 77 a lot of changes were suggested to improve the text.  
   Answer: The suggestions were accepted.

5- On page 4, line 81, change “research assistants” by “RAs”, and also elsewhere.  
   Answer: It was changed.

6- On page 4, line 82, change “and had” by “that may have”.  
   Answer: It was changed.

7- On page 4, line 84; insert the word “such”.  
   Answer: The change was done.

8- On page 5, line 111; remove the word “in”.  
   Answer: It was removed.

9- On page 5, lines 119 and 120, correct the font size.  
   Answer: It was corrected.

10- On page 6, line 126 and 127; change the number “2” by the word “two”.  
    Answer: It was changed.
11- On page 6, lines 127 and 128, insert the word “that” after “foods” and “nutrients”.
   Answer: The suggestion was accepted.

12- On page 6, line 129; change the number “1” by the word “one”.
   Answer: It was changed.

13- On page 7, lines 159, 167 and 171; begin the word “Table” with upper case letter.
   Answer: It was corrected.

14- On page 7, line 164, change “of” by “for” before the words lactating women and reproductive-age women respectively.
   Answer: It was changed.

15- On page 7, lines 164 and 165, insert the word “it” before the word “was”.
   Answer: It was inserted.

16- On page 7, line 168, change the word “among” by “between”.
   Answer: It was changed.

17- On page 8, line 188, change the word “makes” by “enables”.
   Answer: It was changed.

18- On page 8, line 191, insert the word “by” before the word “changed”.
   Answer: It was inserted.

19- On page 8, lines 193 and 194, change “in relation to” by “from those that comprise a”.
   Answer: It was changed.

20- On page 8, line 197, insert the word “the” before “diet” and put the word “nutrient” in plural.
   Answer: The suggestions were accepted.

21- On page 8, line 211, change the word “once” by “as” and the word “will” by “would”.
   Answer: They were changed.

22- On page 8, line 212, change “the pre pregnancy period” by “preconception”.
   Answer: It was changed.

23- On page 9, line 227, change “3rd” by “third.”
   Answer: It was changed.
24- On page 9, lines 232-234, it was suggested to rewrite the selected part of text.
   Answer: It was rewritten.

25- On page 9, line 236, remove the word “the”.
   Answer: It was removed.

26- On page 9, lines 236-237, insert the word “an” before “excessive” and the word “the” before “overall”.
   Answer: It was inserted.

27- On page 9, line 240, insert “that are” before “nutrients”.
   Answer: The suggestion was accepted.

28- On page 9, line 241, change “this may explain” by “this may in part explain”.
   Answer: It was changed.

29- On page 9, line 245, insert the article “a” before the word “recent”.
   Answer: It was inserted.

30- On page 10, line 254, the model used needs to be specified.
   Answer: It was specified.

31- On page 10, line 257, change the word “other” by “another” and the word “with” by “using”.
   Answer: The changes were done.

32- On page 10, line 258, insert the word “occurred” before “however”.
   Answer: It was inserted.

33- On page 10, line 272, delete “and also women”.
   Answer: It was deleted.

34- On page 11, line 283, change “during the life cycle” by “as women make the transition into pregnancy and lactation”.
   Answer: The suggestion was accepted.

35- On page 11, line 285, change “during pregnancy and lactation” by “at this time of the life cycle”.
   Answer: It was changed.